\begin{table}%t4
\caption{\label{table:parameters}Specifications for the different samples\tablefootmark{b}.}
\par
%\centering
\par

\begin{tabular}{c c c c c c c c c c c c c c c c c c}     
\hline\hline \noalign{\smallskip}              
HIP & HD & ${T}_{{\rm eff}}$ & $\sigma({T}_{{\rm eff}})$ & $\log g$ & $\sigma(\log g)$ & [Fe/H] &
$\sigma({\rm [Fe/H]})$ & $\log\epsilon_{\rm Li}$ (LTE)& $\log\epsilon_{\rm Li}$ (NLTE)&
$\sigma(\log\epsilon_{\rm Li})$ & Mass & $\sigma(m)$ & Age & $\sigma(\tau)$ & Source($\tau$) & Parameters & Planets\\
\hline
Sun   & Sun    & 5777 &   -- & 4.44 &  --    &  0.000 &   --   & 1.03 & 1.07 &  0.04 & 1.00 & 0.01 &  4.5 & 0.5 & iso & --   & yes\\
348   & 225194 & 5777 &  40 & 4.41 & 0.07  & --0.130 & 0.024 & 1.08 & 1.11 &  0.05 & 0.97 & 0.01 &  6.2 & 1.5 & iso & R09 & no\\
996   & 804    & 5860 &  41 & 4.38 & 0.07  &  0.000 & 0.022 & 1.56 & 1.59 &  0.05 & 1.03 & 0.01 &  5.1 & 1.1 & iso & R09 & no\\
1499  & 1461   & 5756 &  44 & 4.37 & 0.05  &  0.189 & 0.015 & 0.89 & 0.95 & --1.00 & 1.06 & 0.01 &  4.8 & 0.6 & iso & R09+V05+LH06+T07+S08 & yes\\
2131  & 236416 & 5720 &  41 & 4.38 & 0.07  & --0.210 & 0.026 & 0.10 & 0.13 & --1.00 & 0.92 & 0.01 &  8.9 & 1.3 & iso & R09 & no\\
2894  & --      & 5820 &  44 & 4.54 & 0.07  & --0.030 & 0.025 & 1.81 & 1.84 &  0.05 & 1.02 & 0.01 &  1.6 & 0.8 & iso & R09 & no\\
4909  & 6204   & 5836 &  54 & 4.44 & 0.07  &  0.020 & 0.024 & 2.42 & 2.43 &  0.06 & 1.03 & 0.01 &  0.7 & 0.2 & x-ray & R09+G09 & no\\
5134  & 6470   & 5779 &  38 & 4.49 & 0.07  & --0.190 & 0.023 & 1.81 & 1.84 &  0.05 & 0.97 & 0.01 &  4.1 & 1.9 & iso & R09 & no\\
6407  & 8291   & 5787 &  25 & 4.47 & 0.03  & --0.090 & 0.011 & 1.74 & 1.77 &  0.03 & 0.99 & 0.01 &  3.8 & 0.8 & iso & R09 & no\\
7245  & 9446   & 5843 &  47 & 4.53 & 0.07  &  0.100 & 0.023 & 1.87 & 1.91 &  0.06 & 1.07 & 0.01 &  1.3 & 0.7 & iso & R09 & yes\\
8507  & 11195  & 5720 &  55 & 4.44 & 0.08  & --0.080 & 0.026 & 1.58 & 1.62 &  0.07 & 0.96 & 0.01 &  5.6 & 2.0 & iso & R09 & no\\
8841  & --      & 5676 &  45 & 4.50 & 0.06  & --0.120 & 0.021 & 0.19 & 0.23 & --1.00 & 0.95 & 0.01 &  4.3 & 1.8 & iso & R09 & no\\
9349  & 12264  & 5825 &  28 & 4.49 & 0.06  &  0.010 & 0.017 & 2.03 & 2.06 &  0.04 & 1.03 & 0.01 &  2.5 & 1.2 & iso & R09+T07 & no\\
10710 & --      & 5817 &  43 & 4.39 & 0.06  & --0.130 & 0.022 & 1.90 & 1.92 &  0.05 & 0.98 & 0.01 &  6.5 & 1.1 & iso & R09 & no\\
11072 & 14802  & 5897 &  84 & 4.01 & 0.058 & --0.037 & 0.057 & 2.42 & 2.42 &  0.02 & 1.15 & 0.02 &  0.7 & 0.1 & rot & TW+B07 & no\\
11728 & 15632  & 5738 &  30 & 4.37 & 0.05  &  0.045 & 0.019 & 1.29 & 1.34 &  0.04 & 0.99 & 0.01 &  6.9 & 0.8 & iso & R09+T07 & no\\
11915 & 16008  & 5793 &  43 & 4.45 & 0.06  & --0.050 & 0.021 & 1.69 & 1.72 &  0.05 & 1.00 & 0.01 &  4.3 & 1.6 & iso & R09 & no\\
12186 & 16417  & 5812 &  34 & 4.09 & 0.05  &  0.094 & 0.04  & 1.70 & 1.74 &  0.07 & 1.11 & 0.01 &  7.0 & 0.4 & iso & TW+V05+S08 & yes\\
14614 & 19518  & 5803 &  28 & 4.47 & 0.03  & --0.104 & 0.016 & 1.59 & 1.62 &  0.03 & 0.99 & 0.01 &  3.6 & 0.9 & iso & R09+T07+TW & no\\
14632 & 19373  & 6026 &  42 & 4.28 & 0.05  &  0.136 & 0.019 & 2.46 & 2.47 &  0.07 & 1.17 & 0.01 &  3.7 & 0.3 & iso & R09+V05 & unknown\\
15457 & 20630  & 5771 &  65 & 4.56 & 0.016 &  0.078 & 0.041 & 2.18 & 2.22 &  0.03 & 1.02 & 0.01 &  0.5 & 0.0 & rot & TW+V05 & unknown\\
18261 & 24552  & 5891 &  34 & 4.44 & 0.05  &  0.002 & 0.016 & 2.27 & 2.28 &  0.04 & 1.05 & 0.01 &  3.1 & 1.1 & iso & R09+T07 & no\\
22263 & 30495  & 5826 &  48 & 4.54 & 0.012 &  0.005 & 0.029 & 2.45 & 2.46 &  0.02 & 1.02 & 0.01 &  0.5 & 0.1 & rot & TW+V05 & unknown\\
22528 & --      & 5683 &  52 & 4.33 & 0.10  & --0.350 & 0.035 & 0.46 & 0.49 & --1.00 & 0.87 & 0.01 & 12.2 & 1.5 & iso & R09 & no\\
23835 & 32923  & 5723 &  33 & 4.16 & 0.05  & --0.184 & 0.017 & 1.66 & 1.69 &  0.05 & 0.96 & 0.01 & 10.7 & 0.4 & iso & R09+V05 & unknown\\
25670 & 36152  & 5755 &  37 & 4.38 & 0.05  &  0.071 & 0.017 & 1.24 & 1.29 &  0.05 & 1.01 & 0.01 &  6.0 & 0.9 & iso & R09+T07 & unknown\\
28336 & 40620  & 5713 &  61 & 4.53 & 0.08  & --0.170 & 0.027 & 1.19 & 1.23 &  0.07 & 0.95 & 0.01 &  4.0 & 1.8 & iso & R09 & no\\
29525 & 42807  & 5715 &  61 & 4.41 & 0.037 & --0.005 & 0.036 & 1.99 & 2.03 &  0.02 & 0.97 & 0.01 &  0.4 & 0.0 & rot & TW+B07 & no\\
30037 & 45021  & 5690 &  30 & 4.42 & 0.06  &  0.05  & 0.03  & 0.66 & 0.71 & --1.00 & 0.98 & 0.01 &  5.9 & 1.4 & iso & M09 & no\\
30502 & 45346  & 5745 &  25 & 4.47 & 0.05  & --0.01  & 0.02  & 0.95 & 0.99 & --1.00 & 1.00 & 0.01 &  3.8 & 1.4 & iso & M09 & no\\
36512 & 59711  & 5740 &  15 & 4.50 & 0.03  & --0.092 & 0.02  & 1.27 & 1.31 &  0.04 & 0.99 & 0.01 &  2.7 & 1.0 & iso & M09+T07+S08 & unknown\\
38072 & 63487  & 5839 &  68 & 4.53 & 0.11  &  0.060 & 0.037 & 1.67 & 1.71 &  0.08 & 1.05 & 0.01 &  2.4 & 1.2 & iso & R09 & no\\
38228 & 63433  & 5693 &  58 & 4.52 & 0.07  &  0.007 & 0.025 & 2.51 & 2.53 &  0.07 & 0.99 & 0.01 &  0.3 & 0.0 & rot & R09+V05, $\tau$ from G00& unknown\\
39748 & 67578  & 5835 &  30 & 4.48 & 0.06  & --0.20  & 0.03  & 2.16 & 2.17 &  0.03 & 0.98 & 0.01 &  3.8 & 1.6 & iso & M09 & no\\
41317 & 71334  & 5724 &  15 & 4.46 & 0.03  & --0.044 & 0.02  & 0.92 & 0.96 & --1.00 & 0.98 & 0.01 &  4.7 & 0.8 & iso & M09+V05+S08 & unknown\\
42438 & 72905  & 5864 &  47 & 4.46 & 0.09  & --0.052 & 0.026 & 2.89 & 2.85 &  0.07 & 1.02 & 0.01 &  0.2 & 0.0 & rot & R09+TW & no\\
43190 & 75288  & 5775 &  30 & 4.37 & 0.06  &  0.12  & 0.03  & 0.65 & 0.70 & --1.00 & 1.04 & 0.01 &  5.3 & 0.9 & iso & M09 & no\\
44324 & 77006  & 5934 &  49 & 4.51 & 0.06  & --0.020 & 0.019 & 2.41 & 2.41 &  0.06 & 1.07 & 0.01 &  1.5 & 0.6 & iso & R09+T07& no\\
44713 & 78429  & 5784 &  35 & 4.36 & 0.027 &  0.096 & 0.024 & 0.91 & 0.96 &  0.10 & 1.03 & 0.01 &  5.8 & 0.6 & iso & TW + VF05 + S08 & unknown\\
44935 & 78534  & 5800 &  25 & 4.41 & 0.05  &  0.07  & 0.02  & 0.95 & 0.99 & --1.00 & 1.03 & 0.01 &  4.6 & 1.0 & iso & M09 & no\\
44997 & 78660  & 5782 &  29 & 4.52 & 0.04  &  0.033 & 0.02  & 0.99 & 1.03 &  0.06 & 1.03 & 0.01 &  1.5 & 0.5 & iso & M09 + T07 & no\\
46066 & 80533  & 5709 &  65 & 4.49 & 0.12  & --0.070 & 0.039 & 1.37 & 1.41 &  0.08 & 0.96 & 0.01 &  5.2 & 2.3 & iso & R09 & no\\
46126 & 81700  & 5890 &  30 & 4.48 & 0.06  &  0.14  & 0.03  & 2.17 & 2.20 &  0.03 & 1.09 & 0.01 &  1.7 & 0.8 & iso & M09 & no\\
47990 & 84705  & 5910 &  40 & 4.53 & 0.08  &  0.12  & 0.03  & 2.24 & 2.27 &  0.02 & 1.10 & 0.01 &  0.8 & 0.4 & iso & M09 & no\\
49572 & --      & 5831 &  52 & 4.33 & 0.06  &  0.010 & 0.021 & 1.32 & 1.35 &  0.06 & 1.02 & 0.01 &  6.6 & 0.8 & iso & R09 & no\\
49756 & 88072  & 5804 &  52 & 4.45 & 0.07  &  0.041 & 0.023 & 1.61 & 1.65 &  0.07 & 1.03 & 0.01 &  3.6 & 1.6 & iso & R09+V05+T07 & unknown\\
50826 & --      & 5725 &  30 & 4.47 & 0.06  & --0.28  & 0.03  & 0.95 & 0.98 &  0.08 & 0.92 & 0.01 &  6.1 & 1.8 & iso & M09 & no\\
51258 & 90722  & 5720 &  25 & 4.23 & 0.05  &  0.360 & 0.03  & 0.72 & 0.80 & --1.00 & 1.17 & 0.02 &  5.1 & 0.2 & iso & M09+V05& unknown\\
52040 & 91909  & 5785 &  44 & 4.51 & 0.06  & --0.090 & 0.021 & 1.69 & 1.72 &  0.05 & 0.99 & 0.01 &  3.0 & 1.2 & iso & R09 & no\\
\hline
52137 & 92074  & 5842 &  69 & 4.56 & 0.08  &  0.070 & 0.026 & 2.25 & 2.28 &  0.08 & 1.06 & 0.01 &  0.9 & 0.5 & iso & R09 & no\\
53721 & 95128  & 5916 &  53 & 4.48 & 0.015 &  0.027 & 0.038 & 1.79 & 1.81 &  0.02 & 1.07 & 0.01 &  1.0 & 0.5 & iso + rot &  TW+V05+S05+M08 & yes\\
54102 & 96116  & 5870 &  30 & 4.51 & 0.06  &  0.04  & 0.03  & 2.20 & 2.22 &  0.03 & 1.06 & 0.01 &  1.6 & 0.7 & iso & M09 & no\\
55409 & 98649  & 5760 &  25 & 4.52 & 0.05  & --0.01  & 0.02  & 0.84 & 0.88 &  0.07 & 1.01 & 0.01 &  2.1 & 0.8 & iso & M09 & no\\
55459 & 98618  & 5838 &  21 & 4.42 & 0.03  &  0.038 & 0.012 & 1.53 & 1.56 &  0.04 & 1.03 & 0.01 &  4.1 & 0.7 & iso & R09+V05+M06+T07 & unknown\\
56948 & 101364 & 5795 &  23 & 4.43 & 0.03  &  0.023 & 0.014 & 1.30 & 1.34 &  0.04 & 1.01 & 0.01 &  4.4 & 0.7 & iso & R09+MR07+T09 & unknown\\
56997 & 101501 & 5559 &  65 & 4.53 & 0.08  & --0.030 & 0.027 & 1.08 & 1.14 &  0.08 & 0.94 & 0.01 &  4.6 & 2.1 & iso & R09+V05 & unknown\\
57291 & 102117 & 5690 &  22 & 4.30 & 0.04  &  0.304 & 0.03  & 0.73 & 0.81 & --1.00 & 1.11 & 0.01 &  5.6 & 0.2 & iso & M09+V05+S08& yes\\
59357 & 105779 & 5810 &  30 & 4.45 & 0.06  & --0.24  & 0.03  & 1.70 & 1.72 &  0.04 & 0.95 & 0.01 &  5.5 & 1.5 & iso & M09 & no\\
59610 & 106252 & 5899 &  62 & 4.34 & 0.041 & --0.034 & 0.041 & 1.71 & 1.73 &  0.04 & 1.04 & 0.01 &  5.7 & 0.7 & iso & TW+V05 & yes\\
60081 & 107148 & 5811 &  21 & 4.38 & 0.04  &  0.315 & 0.03  & 1.33 & 1.40 &  0.06 & 1.12 & 0.01 &  3.4 & 0.5 & iso & M09+V05+S08 & yes\\
60314 & 107633 & 5874 &  72 & 4.52 & 0.10  &  0.110 & 0.033 & 1.85 & 1.89 &  0.09 & 1.07 & 0.01 &  1.9 & 0.9 & iso & R09 & no\\
60370  & 107692 & 5897 &  25 & 4.46 & 0.05  &  0.171 & 0.03  & 2.31 & 2.34 &  0.02 & 1.11 & 0.01 &  1.6 & 0.8 & iso & M09+V05 & unknown\\
60653  & 108204 & 5725 &  30 & 4.38 & 0.06  & --0.29  & 0.03  & 0.92 & 0.95 & --1.00 & 0.90 & 0.01 &  9.6 & 1.2 & iso & M09 & no\\
62175  & 110869 & 5849 &  51 & 4.43 & 0.06  &  0.140 & 0.021 & 1.91 & 1.95 &  0.06 & 1.08 & 0.01 &  2.9 & 1.2 & iso & R09+T07 & no\\
64150  & 114174 & 5755 &  41 & 4.39 & 0.05  &  0.056 & 0.016 & 0.76 & 0.81 & --1.00 & 1.00 & 0.01 &  5.9 & 1.0 & iso & R09+V05+T07 & unknown\\
64497  & 114826 & 5860 & 110 & 4.56 & 0.11  &  0.120 & 0.037 & 2.60 & 2.61 &  0.13 & 1.07 & 0.02 &  1.3 & 0.8 & iso & R09 & no\\
64713  & 115169 & 5815 &  25 & 4.52 & 0.05  & --0.01  & 0.02  & 1.43 & 1.46 &  0.04 & 1.03 & 0.01 &  1.7 & 0.8 & iso & M09 & no\\
64794  & 115382 & 5743 &  61 & 4.33 & 0.08  & --0.100 & 0.027 & 0.61 & 0.65 & --1.00 & 0.96 & 0.01 &  8.8 & 1.0 & iso & R09 & no\\
64993  & 115739 & 5875 &  30 & 4.56 & 0.06  &  0.09  & 0.03  & 2.30 & 2.32 &  0.03 & 1.08 & 0.01 &  0.9 & 0.5 & iso & M09 & no\\
66618  & 118475 & 5951 &  25 & 4.35 & 0.05  &  0.135 & 0.03  & 0.96 & 0.99 & --1.00 & 1.12 & 0.01 &  3.6 & 0.5 & iso & M09+V05 & unknown\\
66885  & 119205 & 5685 &  30 & 4.48 & 0.06  & --0.38  & 0.03  & 0.77 & 0.80 & --1.00 & 0.88 & 0.01 &  7.4 & 1.9 & iso & M09 & no\\
69063  & 123152 & 5670 &  30 & 4.31 & 0.06  & --0.45  & 0.03  & 0.74 & 0.76 & --1.00 & 0.83 & 0.01 & 14.1 & 0.7 & iso & M09 & no\\
71683  & 128620 & 5840 &  22 & 4.33 & 0.04  &  0.228 & 0.03  & 1.52 & 1.57 &  0.03 & 1.11 & 0.01 &  4.3 & 0.3 & iso & M09+V05+PM08& unknown\\
72659  & 131156 & 5517 &  67 & 4.56 & 0.09  & --0.117 & 0.033 & 2.44 & 2.47 &  0.10 & 0.91 & 0.01 &  0.2 & 0.0 & rot & R09+V05 & unknown\\
73815  & 133600 & 5803 &  33 & 4.34 & 0.05  &  0.020 & 0.016 & 0.98 & 1.02 & --1.00 & 1.01 & 0.01 &  6.8 & 0.6 & iso & R09+MR07 & no\\
74341  & 134902 & 5853 &  57 & 4.51 & 0.08  &  0.090 & 0.026 & 1.44 & 1.48 &  0.07 & 1.07 & 0.01 &  1.9 & 0.9 & iso & R09 & no\\
74389  & 134664 & 5859 &  24 & 4.48 & 0.04  &  0.105 & 0.03  & 2.08 & 2.12 &  0.03 & 1.07 & 0.01 &  1.6 & 0.7 & iso & M09+S08 & unknown\\
75923  & 138159 & 5775 &  25 & 4.56 & 0.05  & --0.02  & 0.02  & 2.21 & 2.24 &  0.04 & 1.02 & 0.01 &  0.9 & 0.2 & iso & M09 & no\\
77052  & 140538 & 5697 &  33 & 4.54 & 0.023 &  0.035 & 0.023 & 1.49 & 1.54 &  0.01 & 1.01 & 0.01 &  0.6 & 0.4 & iso & TW+V05 & unknown\\
77466  & --	& 5700 &  56 & 4.40 & 0.09  & --0.280 & 0.028 & 0.43 & 0.46 & --1.00 & 0.90 & 0.01 &  9.3 & 1.9 & iso & R09 & no\\
77740  & 141937 & 5900 &  19 & 4.45 & 0.04  &  0.125 & 0.03  & 2.38 & 2.40 &  0.02 & 1.09 & 0.01 &  1.3 & 0.9 & iso + rot & M09+M05+S05+V05+LH06+S08 & yes\\
77883  & 142331 & 5695 &  25 & 4.39 & 0.05  &  0.04  & 0.02  & 0.75 & 0.80 & --1.00 & 0.98 & 0.01 &  7.0 & 0.9 & iso & M09 & no\\
78028  & --	& 5879 &  98 & 4.57 & 0.12  & --0.030 & 0.041 & 1.79 & 1.81 &  0.11 & 1.04 & 0.02 &  1.8 & 1.0 & iso & R09 & no\\
78680  & 144270 & 5923 &  67 & 4.57 & 0.08  & --0.000 & 0.027 & 2.61 & 2.61 &  0.08 & 1.06 & 0.01 &  0.6 & 0.4 & iso & R09 & no\\
79186  & 145514 & 5709 &  48 & 4.27 & 0.08  & --0.120 & 0.024 & 0.29 & 0.33 & --1.00 & 0.95 & 0.01 & 10.3 & 0.7 & iso & R09 & no\\
79304  & 145478 & 5945 &  30 & 4.53 & 0.06  &  0.11  & 0.03  & 2.09 & 2.12 &  0.03 & 1.11 & 0.01 &  0.4 & 0.2 & iso & M09 & no\\
79578  & 145825 & 5860 &  33 & 4.53 & 0.07  &  0.072 & 0.03  & 2.05 & 2.08 &  0.03 & 1.07 & 0.01 &  1.3 & 0.6 & iso & M09+V05 & unknown\\
79672  & 146233 & 5822 &   9 & 4.45 & 0.02  &  0.051 & 0.02  & 1.55 & 1.59 &  0.04 & 1.04 & 0.01 &  3.1 & 0.5 & iso & M09+TW+S08& unknown\\
80337  & 147513 & 5881 &  33 & 4.53 & 0.024 &  0.033 & 0.022 & 2.01 & 2.04 &  0.02 & 1.06 & 0.03 &  0.6 & 0.1 & rot & TW  (S$^4$N+HARPS)+V05+S08+B07 & yes\\
81512  & --	& 5790 &  58 & 4.46 & 0.07  & --0.020 & 0.025 & 0.89 & 0.92 & --1.00 & 1.01 & 0.01 &  4.0 & 1.7 & iso & R09 & no\\
82853  & 150027 & 5640 &  30 & 4.21 & 0.06  & --0.18  & 0.03  & 0.60 & 0.64 & --1.00 & 0.92 & 0.01 & 12.5 & 0.5 & iso & M09 & no\\
83601  & 154417 & 6071 &  43 & 4.38 & 0.08  &  0.048 & 0.028 & 2.82 & 2.79 &  0.06 & 1.13 & 0.01 &  2.4 & 1.1 & iso & R09+V05 & unknown\\
83707  & 152441 & 5880 &  30 & 4.45 & 0.06  &  0.15  & 0.03  & 1.83 & 1.87 &  0.03 & 1.10 & 0.01 &  2.3 & 1.0 & iso & M09 & no\\
85042  & 157347 & 5692 &  37 & 4.39 & 0.022 &  0.037 & 0.026 & 0.56 & 0.61 & --1.00 & 0.98 & 0.01 &  7.2 & 0.5 & iso & TW+V05+S08 & unknown\\
85272  & 156922 & 5700 &  30 & 4.42 & 0.06  & --0.34  & 0.03  & 0.58 & 0.61 & --1.00 & 0.88 & 0.01 &  9.4 & 1.4 & iso & M09 & no\\
85285  & 157691 & 5730 &  30 & 4.43 & 0.06  & --0.39  & 0.03  & 0.71 & 0.73 & --1.00 & 0.88 & 0.01 &  8.8 & 1.5 & iso & M09 & no\\
\hline
86796  & 160691 & 5809 &  22 & 4.28 & 0.04  &  0.298 & 0.03  & 1.06 & 1.13 &  0.07 & 1.15 & 0.01 &  4.6 & 0.2 & iso & M09+V05+S08& yes\\
88194  & 164595 & 5735 &  21 & 4.40 & 0.03  & --0.071 & 0.010 & 0.98 & 1.02 &  0.04 & 0.96 & 0.01 &  7.3 & 0.6 & iso & R09+V05+T07 & unknown\\
88427  & --	& 5810 &  57 & 4.42 & 0.07  & --0.160 & 0.025 & 0.85 & 0.87 & --1.00 & 0.97 & 0.01 &  5.7 & 1.5 & iso & R09 & no\\
89162  & 165357 & 5835 &  30 & 4.32 & 0.06  &  0.07  & 0.03  & 1.35 & 1.39 &  0.09 & 1.05 & 0.01 &  6.0 & 0.6 & iso & M09 & no\\
89443  & 238838 & 5796 &  73 & 4.48 & 0.12  & --0.020 & 0.038 & 1.06 & 1.09 & --1.00 & 1.01 & 0.01 &  4.4 & 1.9 & iso & R09 & no\\
89650  & 167060 & 5855 &  25 & 4.48 & 0.05  &  0.02  & 0.02  & 1.03 & 1.06 &  0.07 & 1.05 & 0.01 &  2.2 & 1.0 & iso & M09 & no\\
91332  & 171918 & 5775 &  25 & 4.20 & 0.05  &  0.206 & 0.03  & 1.40 & 1.46 &  0.09 & 1.12 & 0.02 &  6.4 & 0.7 & iso & M09+V05 & unknown\\
96402  & 184768 & 5713 &  49 & 4.33 & 0.032 & --0.029 & 0.030 & 0.64 & 0.68 & --1.00 & 0.97 & 0.01 &  8.7 & 0.7 & iso & TW+T07 & no\\
96901  & 186427 & 5737 &  28 & 4.34 & 0.04  &  0.055 & 0.016 & 1.38 & 1.42 & --1.00 & 1.00 & 0.01 &  7.5 & 0.5 & iso & R09+V05+LH06+T07 & yes\\
96895  & 186408 & 5808 &  39 & 4.33 & 0.05  &  0.097 & 0.020 & 0.81 & 0.86 &  0.06 & 1.05 & 0.01 &  6.0 & 0.6 & iso & R09+V05+LH06 & unknown\\
100963 & 195034 & 5802 &  17 & 4.45 & 0.03  &  0.008 & 0.013 & 1.72 & 1.75 &  0.05 & 1.02 & 0.01 &  3.6 & 0.8 & iso & R09+T07+T09 & no\\
100970 & 195019 & 5823 &  40 & 4.23 & 0.026 &  0.083 & 0.025 & 1.39 & 1.43 &  0.10 & 1.06 & 0.01 &  7.2 & 0.4 & iso & TW+V05 & yes\\
102152 & 197027 & 5737 &  47 & 4.35 & 0.06  & --0.010 & 0.022 & 0.65 & 0.69 & --1.00 & 0.98 & 0.01 &  7.7 & 0.9 & iso & R09+M09 & no\\
104504 & 201422 & 5836 &  48 & 4.50 & 0.06  & --0.160 & 0.022 & 2.42 & 2.42 &  0.06 & 1.00 & 0.02 &  3.0 & 1.4 & iso & R09  & no\\
107350 & 206860 & 6015 &  50 & 4.48 & 0.07  & --0.020 & 0.019 & 2.91 & 2.87 &  0.07 & 1.09 & 0.01 &  0.2 & 0.0 & rot & R09+V05 & unknown\\
108708 & 209096 & 5875 &  51 & 4.51 & 0.07  &  0.150 & 0.024 & 2.42 & 2.44 &  0.06 & 1.10 & 0.01 &  1.3 & 0.6 & iso & R09 & no\\
108996 & 209562 & 5838 &  56 & 4.50 & 0.08  &  0.060 & 0.027 & 2.40 & 2.42 &  0.07 & 1.05 & 0.01 &  2.3 & 1.1 & iso & R09 & no\\
109110 & 209779 & 5817 &  60 & 4.46 & 0.033 &  0.062 & 0.030 & 2.51 & 2.52 &  0.01 & 1.04 & 0.01 &  0.7 & 0.1 & rot & TW+V05+T07+B07 & unknown\\
109931 & --      & 5739 &  74 & 4.29 & 0.08  &  0.040 & 0.026 & 0.95 & 1.00 & --1.00 & 1.00 & 0.01 &  8.2 & 0.9 & iso & R09 & no\\
113357 & 217014 & 5803 &  47 & 4.38 & 0.05  &  0.221 & 0.017 & 1.36 & 1.42 &  0.07 & 1.09 & 0.01 &  3.9 & 0.7 & iso & R09+V05+LH06 & yes\\
115604 & --      & 5821 &  51 & 4.43 & 0.06  &  0.140 & 0.019 & 0.85 & 0.90 & --1.00 & 1.07 & 0.01 &  3.1 & 1.2 & iso & R09 & no\\
118159 & 224448 & 5905 &  44 & 4.55 & 0.07  & --0.010 & 0.022 & 2.69 & 2.68 &  0.05 & 1.06 & 0.01 &  0.8 & 0.4 & iso & R09 & no\\    
\hline				  		\end{tabular} 					\tablefoot{\tablefoottext{b} {The abbreviations of the sources in Table~\ref{table:parameters} are the following: R09, M09 and TW are from \cite{Ramirez-Melendez-Asplund-09}, \cite{Melendez-09a},
\cite{Melendez-09b} and this work, as~before. M06~is \cite{Melendez-Dodds-Eden-Robles-06}, V05 is \cite{Valenti-Fischer-05}, S08 is \cite{Sousa-08}, B07 \cite{Barnes-07}, LH06 \cite{Luck-Heiter-06}, PM08 \cite{PortodeMello-Lyra-Keller-08}, T07 and T09 are \cite{Takeda-07} and \cite{Takeda-Tajitsu-09}, respectively, G00 is \cite{Gaidos-Henry-Henry-00} and MR07 is \cite{Melendez-Ramirez-07}. S05 is \cite{Saffe-Gomez-Chavero-05}, G09 is \cite{Guinan-Engle-09}, and M08 \cite{Mamajek-Hillenbrand-08}. The $-1$ in $\sigma(\log\epsilon_{\rm Li})$ denotes upper limits.}}
\end{table}